Author's response to reviews

Title: Incidental findings on MRI scans of patients presenting with audiovestibular symptoms

Authors:

Vasileios Papanikolaou (vspap@hotmail.com)
Mohamed H Khan (mhkhan@hotmail.com)
Ivan J Keogh (ijkeogh@hse.ie)

Version: 4 Date: 3 June 2010

Author's response to reviews: see over
Dear Madam/Sir,

Thank you for accepting our paper for publication and for your comments. I have revised our text as per your instructions in your mail dated 27th May 2010. Please find a detailed description of the changes below:

**Major revisions**
- **Author list:** the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. The preferred style is First name Initial Last name (e.g. Joe F Bloggs). Please also remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submission system. 

Appropriate corrections have been made to the submitted manuscript

- **Affiliations:** Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Appropriate corrections have been made to the submitted manuscript

- **Abstract:** please ensure that the Abstract in the manuscript matches exactly what is in the submission system. At present they are not the same.

Appropriate corrections have been made updating the online version

- **Authors' contributions:** Please reformat, using the following format as an example, including using initials to refer to each author's contribution and stating that all authors read and approved the final manuscript:

Appropriate corrections have been made to the submitted manuscript

- **Figures:** Some of your figures are quite large; the maximum size for each individual figure is A4. If your figure is and needs to remain larger than this, and cannot be reduced, we recommend you separate it into component parts and upload these separately. Please remember to update the figure titles and references within the text.

Appropriate corrections have been made to the submitted figures with a reduction of size:

- Fig1 from 4217x2037 to 1012x489
- Fig2 from 1671x1988 to 401x477
- Fig3 from 5583x2504 to 1340x601
- Fig4 from 5896x2484 to 1415x596

**Minor revisions**
-Email address: please include email addresses for all the authors on the title page,
Appropriate corrections have been made to the submitted manuscript

-Keywords: Please remove.
Appropriate corrections have been made to the submitted manuscript

-Abbreviations: Please format these in to a paragraph.
Appropriate corrections have been made to the submitted manuscript

-Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance
No further corrections have been made

Sincerely yours,

The authors,

Papanikolaou V, Khan MH, Keogh I